Author's response to reviews

Title: A survey of diet self-efficacy and food intake in students with high and low perceived stress

Authors:

Robyn Nataskin (robyn.nastaskin@ryerson.ca)
Alexandra J Fiocco (afiocco@psych.ryerson.ca)

Version: 5 Date: 2 April 2015

Author's response to reviews:

In response to the reviewer's comment, we have included the word "intake" in the title for Figure 1 and 2.